# Three-dimensional simulation of aesthetic outcome from breast-conserving surgery compared with viewing photographs or standard care: randomized clinical trial

A. R. Godden 🔟 <sup>1,2</sup>, A. Micha<sup>1</sup>, L. M. Wolf<sup>1</sup>, C. Pitches<sup>2</sup>, P. A. Barry<sup>1</sup>, A. A. Khan<sup>3</sup>, K. D. C. Krupa<sup>1</sup>, A. M. Kirby<sup>1,2</sup> and J. E. Rusby<sup>1,2,\*</sup>

<sup>1</sup>Department of Breast Surgery, Royal Marsden NHS Foundation Trust, Sutton, Surrey, UK

<sup>2</sup>Independent patient co-designer, Institute of Cancer Research, Sutton, Surrey, UK

<sup>3</sup>Department of Plastic Surgery, The Royal Marsden NHS Foundation Trust, London, UK

\*Correspondence to: Orchard House, The Royal Marsden Hospital, Sutton, Surrey, SM2 5PT, UK (e-mail: Jennifer.rusby@rmh.nhs.uk)

#### Abstract

**Introduction:** Over half of women with surgically managed breast cancer in the UK undergo breast-conserving treatment (BCT). While photographs are shown prior to reconstructive surgery or complex oncoplastic procedures, standard practice prior to breast conservation is to simply describe the likely aesthetic changes. Patients have expressed the desire for more personalized information about likely appearance after surgery. The hypothesis was that viewing a three-dimensional (3D) simulation improves patients' confidence in knowing their likely aesthetic outcome after surgery.

**Methods:** A randomized, controlled trial of 117 women planning unilateral BCT was undertaken. The randomization was three-way: standard of care (verbal description alone, control group), viewing two-dimensional (2D) photographs, or viewing a 3D simulation before surgery. The primary endpoint was the comparison between groups' median answer on a visual analogue scale (VAS) for the question administered before surgery: 'How confident are you that you know how your breasts are likely to look after treatment?'

**Results:** The median VAS in the control group was 5.2 (i.q.r. 2.6–7.8); 8.0 (i.q.r. 5.7–8.7) for 2D photography, and 8.9 (i.q.r. 8.2–9.5) for 3D simulation. There was a significant difference between groups (P < 0.010) with post-hoc pairwise comparisons demonstrating a statistically significant difference between 3D simulation and both standard care and viewing 2D photographs (P < 0.010 and P = 0.012, respectively).

**Conclusion:** This RCT has demonstrated that women who viewed an individualized 3D simulation of likely aesthetic outcome for BCT were more confident going into surgery than those who received standard care or who were shown 2D photographs of other women. The impact on longer-term satisfaction with outcome remains to be determined. Registration number: NCT03250260 (http://www.clinicaltrials.gov).

This work was presented in part at the Association of Breast Surgery Annual Conference, Glasgow 2019, and was accepted for presentation at The European Breast Cancer Conference, Barcelona 2020.

#### Lay summary

Most women with breast cancer are able to have an operation to remove the cancer while preserving the breast ('lumpectomy'). Whilst cancer control is the most important goal, appearance after surgery has been shown to affect long-term quality of life and is considered when planning treatment. Currently, surgeons simply describe the likely changes in appearance and, for more complex procedures, photographs of other women are shown. Patients themselves have indicated they would like more information regarding the likely changes to their breast after treatment. The authors have developed a way to simulate appearance following lumpectomy and radio-therapy using three-dimensional (3D) photographs. The study invited women undergoing lumpectomy to be assigned at random to one of three groups receiving standard care (discussion), a two-dimensional photograph, or the 3D simulation before their operation. The authors have demonstrated that showing a woman her simulation prior to surgery improves confidence going into treatment.

#### Introduction

Breast cancer is a common and emotive diagnosis with 55176 patients diagnosed in the UK annually  $(2015-2017)^1$ . Of all women with surgically managed breast cancer, over half undergo

breast-conserving surgery (BCS), with 28 500 operations performed annually in the  $UK^2$ . The development of oncoplastic techniques and more effective neoadjuvant chemotherapy enable a larger proportion of women to consider breast conservation.

© The Author(s) 2021. Published by Oxford University Press on behalf of BJS Society Ltd.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

Received: January 12, 2021. Accepted: May 06, 2021

Shared decision making has been a focus of NHS England since 2013 when it took over the Shared Decision-Making Programme from the Quality, Innovation, Productivity and Prevention (QIPP) Right Care Programme<sup>3</sup>. Shared decision making is considered a standard of care in breast cancer. Literature focusing on patients' experience of shared decision making within breast cancer treatment has described reduced levels of stress, improved knowledge and a preference for a personalized decision-making approach<sup>4,5</sup>. Shared decision making has recently been adopted by the General Medical Council as a formal recommendation within the decision-making and consent guidelines published in September 2020<sup>6</sup>.

Three-dimensional (3D) simulation is a visual experience that has been used to provide a personalized approach to care. In the cosmetic surgery industry, particularly within breast and facial surgery, 3D surface imaging (3D-SI) and simulation are widely used to facilitate patient decision making<sup>7,8</sup>. Simulation was reported to be highly reproducible for breast augmentation<sup>9–11</sup>, and proved a useful tool for implant selection<sup>12–15</sup>. Simulation is designed to display complex ideas simply, crossing language and literacy barriers and potentially improving communication in the preoperative planning stage of surgery. 3D simulation may also add value as a tool to improve patient confidence in knowing their likely aesthetic outcome after breast cancer surgery by reducing the gap between patient perception (how they interpret an explanation) and expectation (their visualization of their anticipated result).

The simulation software available uses predefined algorithms to model outcome from aesthetic surgery (implant augmentation, lipo-filling and mastopexy). There is no software currently available to model breast reconstruction or BCS using 3D-SI. Some groups have looked at complex modelling of the outcome of BCS using biomechanics and wound-healing models based on MRI imaging, but these methods involve complex mathematics, are time consuming and expensive, and have not yet been translated to the clinical setting<sup>16-18</sup>.

In many breast units, the standard preoperative preparation for a woman undergoing BCS includes a verbal description of likely aesthetic changes. Often women undergoing breast reconstruction are shown photographs of other women who have had similar operations. Women themselves have explained that looking at other women's postoperative photographs did not always give them a sense of how *they* would look, and some reported that it felt inappropriate and awkward. The concept of using simulation as part of a preoperative discussion within breast cancer surgery was generated by a patient steering group as a desirable area of study.

The hypothesis was that confidence approaching a BCS operation may be increased if a woman has reviewed simulated images of her own appearance, and this may translate to better satisfaction with outcome in terms of satisfaction with the breasts. Conversely, if the simulation gives a woman an artificially high level of expectation then she may be more disappointed than had she not seen it.

The aim of this study was to establish, using a randomized controlled trial design, which preoperative intervention best prepares women for their likely aesthetic outcome after BCS: verbal description alone; viewing others' (2D) photographs; or viewing a 3D simulation of their own likely appearance.

#### Methods

#### Patient selection

This study received NHS Research Ethics Committee and Health Research Authority approval (clinicaltrials.gov identifier NCT03250260). The Royal Marsden Sutton site, where recruitment took place, predominantly serves the local population with symptomatic and screen-detected breast cancer, rather than providing tertiary care for complex cases. Women with early breast cancer without previous breast surgery undergoing unilateral BCT (wide local excision or therapeutic mammoplasty with the expectation of adjuvant radiotherapy) without preoperative expectation of subsequent contralateral symmetrization were eligible.

Potential participants were identified prior to surgery by the clinical team of four consultant oncoplastic breast surgeons. The study was introduced by a member of the clinical team (either breast care nurse or the surgeon during consultation) and the patient information sheet was issued along with an infographic. A telephone call followed from one of two clinical research registrars, and an initial study consultation was arranged (to coincide with preoperative assessment to limit the number of hospital visits). During the recruitment process, a standardized explanation of the study was given to ensure that patients understood that the researchers had equipoise.

Participants were randomized before surgery by the Institute of Cancer Research Clinical Trials Unit. This was separate from the recruitment pathway, ensuring allocation concealment. Stratified randomization with random permuted blocks was used for this trial. The mechanism used to generate the randomization lists was a central, independent, bespoke computer program. Patients were randomized into one of three groups with a 1:1:1 ratio: standard care (verbal description of likely aesthetic changes alone, the control group); viewing 2D photographs of other women who had undergone similar treatment (surgery and radiotherapy) in the past (matched for age, BMI, breast volume and tumour location); or viewing a real-time simulation of an average outcome from BCS using their own 3D-SI. Randomization was stratified by BMI, intention to undergo axillary lymph node dissection and operation type (standard wide local excision or more complex tissue rearrangement). Axillary lymph node dissection and high BMI have previously been shown to be independently associated with poorer patient-reported satisfaction with breasts<sup>19</sup>.

The primary endpoint was the difference between groups' median score on a visual analogue scale (VAS) administered before surgery for the question:

'How confident are you that you know how your breasts are likely to look after treatment?'

It was not possible to blind the investigator to the allocated group at the initial study consultation. However, to reduce bias, the VASs were measured *en masse* at completion of recruitment by one investigator blinded to the randomization group.

#### Consultation process

One of two investigators (A.R.G. or A.M.) conducted the initial study consultation. Written informed consent was obtained. All women completed the baseline BREAST-Q BCS questionnaire<sup>20</sup>, a validated patient-reported outcome tool and underwent 3D image capture (VECTRA XT<sup>®</sup>, Canfield Scientific, New Jersey, USA). Vectra<sup>®</sup> is a 3D photographic image capture system. Six mounted cameras take simultaneous images, which are then integrated into a 3D image viewable on a workstation. The VECTRA XT is easy to use, has a fast capture speed (3.5 ms) and processing speed (80 seconds) and does not require an experienced photographer. Other 3D surface-imaging systems are available. Women in the 2D and 3D groups then reviewed images, either standardized 2D postoperative photographs of other women or their own 3D

simulations, respectively. A standardized verbal description was used with all groups to explain likely aesthetic changes in an attempt to equalize the amount of time spent with a member of the clinical team and reduce the possibility of bias. The final part of the initial consultation for all women was to mark the VAS that was administered in a private room and handed to the investigator in a sealed envelope, marked only with the participant's study number.

#### Secondary endpoints

Follow-up is on-going and secondary endpoints will be reported in due course. Here, they are summarized for context. The secondary aims are to investigate the effectiveness of the simulation compared with the other preparatory methods (randomization groups) using a second VAS administered at 3 and 12 months after BCS to address the question: 'How well do you think the information about how your breasts are likely to look after surgery (discussion, 2D photographs, or 3D simulation) reflects how they actually look today?' The accuracy of the simulation compared with reality (at 1 year after BCS) will be compared using both linear and 3D measures obtained from 3D-SI. Changes over time with regard to both patient-reported outcome measures (BREAST-Q BCS module) and objective measures taken from annual 3D-SIs will be reported out to 5 years after completion of BCS. The study will end when the final patient completes 5 years of follow-up.

#### Simulation model

The simulation method was based on pre-existing software for simulating mastopexy outcome within Mirror<sup>TM</sup> software, and the VECTRA XT<sup>®</sup> capture. An adjustment was made to each patient's preoperative 3D image based upon the average difference for a series of 200 patients who underwent BCS with the best and worst results (65 patients) excluded (135 remained)<sup>21</sup>. Mirror<sup>TM</sup> software is designed to simulate implant augmentation and mastopexy. Although there are many options for manipulating the images, it is not currently possible to change certain objective measures by a predefined amount. Nipple-sternal notch (N-SN) distance can be manipulated in the mastopexy function, so this was used for the simulation. N-SN distance was reduced by 5 per cent and the nipple moved laterally. A circumareolar scar pattern was used to standardize the type of simulation used. No patient-specific information was entered. Example simulations are shown in Fig. 1.

The simulation process was checked using a small number of 3D-SIs from the development series of patients to provide proof of principle: The simulation process was performed on the unoperated breast and compared visually with the treated breast (actual outcome). The accuracy of the simulation to display an average outcome from BCT was deemed acceptable by two reviewers for the purpose of this trial. Natural asymmetry of the breasts is not accounted for in this model, but in the absence of a large set of pre- and postoperative 3D-SI, this was a pragmatic approach.

Prior to showing the simulation to participants randomized to that group it was stressed that the images they were about to see were based on average changes seen in women having similar but not identical treatment, therefore, it was designed to give an idea of the average post-treatment outcome but 'may not' represent exactly how they would look. They were told that they may have an aesthetic outcome that is better or worse than the images they were going to view.

#### Power calculation and statistical analysis

The primary endpoint was based on a visual analogue scale (VAS). A difference of 1.5 or greater between any two groups, was considered clinically significant. This was based on clinician judgement and feedback from patient representatives. With an estimated standard deviation of 2.0 using a 2-tailed t test and a Bonferroni corrected alpha of 0.017 to allow for three comparisons, the study required 39 patients per arm giving a total of 117 patients to have 80 per cent power to detect a difference. Although it was predicted that 15 per cent of patients might be lost due to mastectomy or drop out later, this would occur after completing the VAS and would not, therefore, affect the primary outcome measure and did not need to be accounted for in the power calculation. A Kruskal–Wallis test was used to compare preoperative VAS scores between the three groups with a 5 per cent significance level, with further post-hoc tests to find intergroup statistically significant differences. In presentation of clinicopathological variables, between-group differences were analysed using the one-way ANOVA for continuous variables and the  $\chi^2$  test for categorical variables. BREAST-Q descriptive statics were represented using median (i.q.r.). Between-group differences were analysed using a Kruskal–Wallis test (non-parametric data).

#### **Results**

#### Recruitment

The study opened in May 2017 and the final patient was recruited in October 2019. The consort flow diagram is shown in Fig. 2. Forty-one participants were allocated to the control group, 39 to the 2D-photograph group and 37 to the 3D simulation group. There were no statistically significant differences in stratification factors by randomization group.

#### Demographics

The demographics and clinicopathological data for the study population are shown in *Table* 1. The participants had an average age of 59 years, BMI of  $29 \text{ kg/m}^2$  and were mainly white British, reflecting the local population. The majority of women underwent standard wide local excision (67 per cent) and sentinel lymph node biopsy (79 per cent). Two participants were deemed unfit for surgery after randomization and hence are not included in the histopathological data but are included in the primary endpoint. There were no significant between-group differences for the 'satisfaction with breasts' domain of the preoperative BCT BREAST-Q (P=0.343). Group two (2D photographs) was significantly older than groups one (control) and three (simulation) (P=0.013 and P < 0.001 respectively). No difference was observed between groups one and three (P=0.777).

#### Primary endpoint

The total number of participants for the primary endpoint was 116 because it was not possible to perform a simulation using the Mirror<sup>TM</sup> software for one participant. The final group sizes are uneven due to the number of stratification factors used in the trial. For each possible combination of the factors, there is a separate set of randomized blocks, the aggregated blocks forming a specific 'randomization list' for that type of patient. The patients are allocated 1:1:1 within each list, however, if the number of patients on a list is small the list might not fully complete the blocks of 3 and 6, resulting in a slight imbalance.

The median VAS score was 5.2 (i.q.r. 2.6–7.8) for standard care, 8.0 (i.q.r. 5.7–8.7) for viewing 2D photographs and 8.9 (i.q.r. 8.2–



#### Fig. 1 Example simulations

Top row (participant A) from left to right; preoperative appearance, simulation of postoperative appearance (left breast), actual postoperative appearance at 12 months after treatment. Bottom row (participant B) from left to right; preoperative appearance, simulation of postoperative appearance (right breast), actual appearance at 12 months after treatment

9.5) for women who viewed their own simulation. This was statistically significantly higher in the 3D simulation group than for standard care and 2D photograph groups (P < 0.001 and P = 0.012, respectively). Of note, the women who viewed their simulation showed much less variation in their responses with a tight interquartile range compared with either the control or 2D photograph groups. No statistically significant difference was observed between standard verbal discussion and 2D photographs (P = 0.061) although a trend in favour of 2D photographs was noted (Fig. 3).

#### Discussion

This is the first randomized controlled trial investigating the impact of 3D simulation of appearance on women's preoperative understanding of aesthetic outcome. This study also evaluates the impact of 3D-SI in the provision of information about BCS. Participants who viewed a real-time 3D simulation of their own likely appearance reported a greater level of confidence in their knowledge of their likely aesthetic outcome than the other two groups.

In the cosmetic surgery industry, particularly within breast and facial surgery, 3D-SI and simulation are widely used to facilitate patient decision making<sup>7,8</sup>. In a survey of members of the American Academy of Facial Plastic and Reconstructive Surgery, 63 per cent of surgeons already use simulation as part of their rhinoplasty consultation<sup>22</sup>. Patients appreciate the use of simulation in preoperative decision making consultation for aesthetic surgery with a reported 70 per cent of patients undergoing rhinoplasty stating they would decline surgery in its absence<sup>23</sup>. Patients also report a higher satisfaction with 3D simulation over 2D simulation for rhinoplasty<sup>24</sup>. Persing and colleagues used panel evaluation to examine the accuracy of 3D rhinoplasty simulation using VECTRA and deemed actual aesthetic results to be superior to simulation<sup>23</sup>. The group also concluded that experienced surgeons are necessary to translate the simulation into an achievable plan. Simulation was reported to be highly

reproducible for breast augmentation<sup>9–11</sup>, and proved a useful tool for implant selection<sup>12–15</sup>. Patients found preoperative simulation for breast augmentation helpful and reported satisfaction with their preoperative decisions<sup>14</sup>. It has been shown to be useful for measuring the anticipated volume changes in aesthetic surgery<sup>9,13,25</sup>.

de Runz and co-workers used Crisalix<sup>TM</sup> (Virtual Aesthetics, Crisalix, Lausanne, Switzerland) to simulate breast augmentation for 38 women. Six months after surgery, 66 per cent of the women absolutely agreed that the simulation represented their actual outcome and 24 per cent partially agreed. Ninety-three per cent felt the simulation helped them choose their implant size, and 97 per cent found the simulation useful<sup>26</sup>. Vorstenbosch and colleagues also used Crisalix to simulate breast augmentation and asked an expert panel to comment upon its accuracy compared with postoperative 3D-SI. The results highlighted baseline breast type as an influencing factor for simulation success<sup>11</sup>. The simulation was deemed to predict overly optimistic results for women with ptotic breasts, and the opposite for women with tuberous breasts. The most accurate simulations were for women with symmetrical breasts at baseline<sup>11</sup>. This may also be relevant with Mirror<sup>TM</sup> software as it was difficult and occasionally not possible for the software to perform simulation in women with large breasts, high BMI or grade 3 ptosis (Regnault classification)<sup>27</sup>. These studies were concerning breast augmentation where alteration of breast volume is more predictable (a standardized implant) compared with BCS for cancer where the volume (of tumour) excised is rather less exact. Clinically this may still be pertinent to patient selection for simulation for BCS until a workaround with the software can be written or bespoke simulation becomes normality.

The simplicity of the simulation method was one of the strengths of this study. It provides a rapid simulation of aesthetic outcome (less than 5 minutes), not requiring complicated calculations and could be completed during a clinical consultation. This simulation enabled participants to view an average outcome from BCS on their own breast and provides proof-of-principle



#### Fig. 2 Consort flow diagram

This work was in collaboration with the Royal Marsden Clinical Trials Unit, with randomisation services provide by the Institute of Cancer Research Clinical Trials Statistics Unit

that viewing simulation was superior to current standards of explanation, prior to investing time and money in development of a bespoke simulation process. The simple simulation method may also be viewed as a weakness of the study. Although for the majority of women, the simulation would have been close to the actual outcome, for a proportion it would have demonstrated a better or worse appearance than the participant went on to receive. In order to provide an average result and avoid an overly optimistic or pessimistic example, the simulation was centreweighted and based upon a BCS population who had scored 2 or 3 out of 4 (Harvard cosmesis scale) for aesthetic outcome. Inaccuracy of the simulation may influence subsequent patientreported outcome measures, which will therefore be reported as secondary endpoints.

The development of a bespoke simulation may be required to represent aesthetic outcome better, allowing the surgeon to take individual technical and patient factors into account, such as the size and location of the tumour, scar pattern and potential for symmetrization. The prediction of aesthetic outcome for oncological resections presents additional challenges compared with aesthetic procedures due to the degree of uncertainty with regard to excision volumes and the effect of adjuvant treatment. Simulation may be used as an adjunct to the viewing of 2D photographs in order to give a range of outcomes for patients to avoid setting expectations too high in the initial phases of development.

There is also clinical interest in development of simulation as a tool for shared decision making, as this may help women considering complex oncoplastic breast conservation and breast reconstruction, that is therapeutic mammoplasty with or without symmetrization *versus* unilateral mastectomy and reconstruction or between different types of reconstruction. Knoops and colleagues have developed a machine-learning base framework method using 3D-SI (3DMD, LLC, Atlanta, Georgia, USA) to simulate postoperative appearance for craniofacial surgery with the intended application of more precise surgical planning and outcome evaluation<sup>28</sup>. The use of artificial intelligence may be the

#### Table 1 Demographics and clinicopathological data for the simulation study population

Demographics	Total (n = 117)	Control (n = 41)	2D photographs (n = 39)	3D simulation ( $n = 37$ )	Р
Age* BMI*	59(10) 29(6)	58(9.17) 28(5.65)	64(8.59) 29(6.59)	55(10.46) 28(5.6)	<0.001
Ethnicity	23(0)	20(3.03)	23(0.55)	20(3.0)	0.511
White British	94 (80)	32	31	30	
White other	12 (10)	2	5	5	
Asian	4 (3)	3	1	1	
Black Caribbean	4 (3)	3	0	1	
Mixed white and Asian	1 (1)	0	1	0	
Black African	1 (1)	1	0	Ő	
Black other	1 (1)	0	1	Ő	0 426
Smoking status	1 (1)	0	Ť	Ũ	0.120
Non-smoker	61 (52)	19	21	21	
Fx-smoker	42 (36)	15	15	11	
Smoker	14 (12)	6	3	5	0 749
Cliniconathological information	14(12)	0	2	2	0.745
Laterality of cancer					
Laterality of cancer	63 (54)	22	20	21	
Pight	54 (46)	10	10	16	0 801
Tumour location	J + (10)	15	19	10	0.001
Upper outer quadrant	51 (11)	10	14	19	
Upper outer quadrant	17 (15)	6	6	5	
Upper control	17 (13)	U E	5	2	
Latoral	12 (10)	1	5	2	
Lateral Lower outer augdrapt	11 (9) 9 (7)	1	5	1	
Lower outer quadrant	0 (7) 6 (5)	2	2	1	
Lower miller quadrant	0 (S) E (4)	5	2	0	
Control	) (4) (2)	1	2	2	
Vedial	4 (5)	2	0	2	0.001
	5 (5)	T	0	Z	0.091
Monorma of the mark (normal)*	20(12.20)	00(10 OC)	10(10.00)	22(15.2)	0 000
Maininogram (mm)*	20(12.69)	22(15.20)	10(15.20)	22(15.5)	0.562
$MDL(mm)^*(m = 12)$	10(11.27)	10(9.49)	20(15.79)	19(9.90)	0.010
MRI(IIIII) (n = 13)	26(12.34)	31(14.7)	24(2.83)	29(28.44)	0.941
Wide legal evolution	70 (67)	77	24	2F	
Mammanlasty	76 (07)	27	24	20	0 007
	39 (33)	14	13	12	0.987
	00 (70)	25	2.9	20	
SLINB	92 (79)	35	28	29	
Mana	/ (0)	3	ے ج		
	10 (14)	43	/	0	0 610
	∠ (∠)	U	Ţ	Ţ	0.618
Satisfaction with broasts O score <sup>†</sup>	61 (52 00)	61 (10 07)	EQ (10 71)	E0 (E2 07)	0 2 4 3
Sausiacuon with breasts Q-score	64 (53-82)	b4 (4ŏ−ŏ∠)	Jð (4ð-/ 1)	58 (53-87)	0.343

Values in parentheses are percentages unless indicated otherwise; Between-group differences were analysed using the one-way ANOVA for continuous variables and the  $\chi$ 2 test for categorical variables. \*values are mean(s.d.), <sup>†</sup>values are median (i.q.r.). SLNB, sentinel lymph node biopsy; ALNC, axillary lymph node clearance; TAD, targeted axillary dissection; BCT, breast-conserving treatment.

most accurate, reliable and efficient way to develop simulation for breast reconstruction, although a large library of images would be required in order to capture the diversity of UK oncoplastic breast surgical practice.

A potential area of bias is that more time spent with a clinician is an intervention in itself. To mitigate against this, although the simulation group was compared with relevant alternatives including a verbal description (control group) and the viewing of 2D photographs, every woman had a 3D image capture at baseline (so experienced the technology even if they did not see their own simulation). The number of investigators was minimized at two, and a standard description was used for every participant, regardless of group, to explain the common aesthetic changes observed from BCS prior to randomization. The investigator and patient could not be blinded during the allocated consultation and the potential psychological influence of 'getting to see' their simulation could not be avoided. Nor could the discussion into expected aesthetic outcome between the participant and their surgeon in the clinical environment be controlled, though this happened prior to randomization. The investigators were blinded

to the randomization group during outcome analysis. The primary endpoint was the patient's response on a VAS to a non-validated question about confidence in knowing their likely aesthetic outcome after surgery. There was no alternative validated questionnaire option at the inception of this study.

It is well described that improving preoperative information can positively influence postoperative recovery amongst surgical patients<sup>29-31</sup>. Preoperative counselling in breast cancer patients has been shown to improve social recovery including postoperative aesthetic satisfaction<sup>32</sup>. Involvement in decision making per se, although preferred by many women, is not reported independently to influence psychosocial morbidity, rather the quality of information given seems to be the crucial step in long-term psychosocial adjustment<sup>33</sup>. Within the breastreconstruction population, improved preoperative education has been reported to reduce decisional regret and improve satisfaction<sup>34-36</sup>.

The superiority of 3D simulation in preoperative explanation of aesthetic results is demonstrated in this study. The utility of simulation for BCS in routine clinical practice has yet to be



**Fig. 3** Box and whisker plot demonstrating between-group differences for the primary endpoint VAS in answer to the question administered before surgery: 'How confident are you that you know how your breasts are likely to look after treatment?'

n = 116 (1 failed simulation). X, mean

defined, as has whether managing expectations before surgery using simulation will affect longer-term outcomes such as satisfaction with breasts and, as a result, psychosocial quality of life. With completion of follow-up, the long-term influence of simulating outcome on patients' 'satisfaction with breasts' and quality of life may provide additional rationale for the routine use of simulation in improving the quality of survivorship. The 5-year follow-up has been designed to measure the evolving changes to the post-treatment breast objectively over time. Potential influencing factors such as operation type and progressive fibrosis from radiotherapy will be assessed. This information will be used to refine bespoke simulation and to aid informed preoperative decision making.

In summary, simulation provides an efficient way to share information in the preoperative setting and demonstrates superior results, compared with current methods, when preparing women for their aesthetic outcome. In order to translate simulation from research into actual practice, this needs to be feasible in the clinic. The long-term influence of simulating outcome on patients' 'satisfaction with breasts' and quality of life requires further investigation.

#### Acknowledgements

This paper represents independent research funded by the Pink Ribbon Foundation and by the National Institute for Health Research (NIHR) Biomedical Research Centre at The Royal Marsden NHS Foundation Trust and the Institute of Cancer Research. The views expressed are those of the authors and not necessarily those of the NHS, the NIHR or the Department of Health and Social Care. All procedures performed in studies involving human participants were in accordance with the ethical standards of the institutional and/or national research committee and with the 1964 Helsinki Declaration and its later amendments or comparable ethical standard. The protocol was reviewed by London-Riverside NRES committee Ref: 17/LO/0399. The study is registered on a publicly accessible database, clinicaltrial.gov, NCT03250260. The full protocol can be found at The Royal Marsden Hospital (reference CCR4660).

The authors would like to formally acknowledge the contributions of the participants, and medical photographer Dennis Underwood to this study and to offer posthumous thanks to Carol Pitches, an eloquent voice in patient co-design, marking her contribution to this project and the wider breast surgery research program at the Royal Marsden Hospital.

Conflicts of Interest: The authors have no conflicts of interest to declare.

#### Funding

A.R.G. is funded by a grant from the NIHR Royal Marsden/ Institute of Cancer Research Biomedical Research Centre.

Disclosure. The authors declare no conflict of interest.

#### **References**

- Cancer Research UK. Breast Cancer Survival Statistics. 2020. https://www.cancerresearchuk.org/health-professional/can cer-statistics/statistics-by-cancer-type/breast-cancer/survival# heading-Zero.
- Jeevan R, Cromwell DA, Trivella M, Lawrence G, Kearins O, Pereira J et al. Reoperation rates after breast conserving surgery for breast cancer among women in England: retrospective study of hospital episode statistics. BMJ 2012;345:e4505.
- NHS England. Shared Decision Making. 2020.NHS England » Shared decision making. https://www.england.nhs.uk/shareddecision-making/ (accessed 15 June 2021).
- Serpico V, Liepert AE, Boucher K, Fouts DL, Anderson LA, Pell J et al. The effect of previsit education in breast cancer patients: a study of a shared-decision-making tool. Am Surg 2016;82:259–265.
- 5. Scott AM, Jhanwar SM, Pusic A, McCarthy CM. Shared decisionmaking in women with early stage breast cancer and

implications for long-term health-related quality of life. Value Health 2014;**17**:A651.

- General Medical Council. Decision Making and Consent 2020. https://www.gmc-uk.org/ethical-guidance/ethical-guidancefor-doctors/decision-making-and-consent (accessed 15 June 2021).
- Georgii J, Eder M, Bürger K, Klotz S, Ferstl F, Kovacs L et al. A computational tool for preoperative breast augmentation planning in aesthetic plastic surgery. *IEEE J Biomed Health Inform* 2014;18: 907–919.
- Bottino A, De Simone M, Laurentini A, Sforza C. A new 3-D tool for planning plastic surgery. *IEEE Trans Biomed Eng* 2012;59: 3439–3449.
- Mailey B, Freel A, Wong R, Pointer DT, Khoobehi K. Clinical accuracy and reproducibility of Portrait 3D Surgical Simulation Platform in breast augmentation. Aesthet Surg J 2013;33:84–92.
- Gladilin E, Gabrielova B, Montemurro P, Hedén P. Customized planning of augmentation mammaplasty with silicon implants using three-dimensional optical body scans and biomechanical modeling of soft tissue outcome. *Aesthetic Plast Surg* 2011;35: 494–501.
- Vorstenbosch J, Islur A. Correlation of prediction and actual outcome of three-dimensional simulation in breast augmentation using a cloud-based program. Aesthetic Plast Surg 2017;41: 481–490.
- Creasman CN, Mordaunt D, Liolios T, Chiu C, Gabriel A, Maxwell GP. Four-dimensional breast imaging, part II: clinical implementation and validation of a computer imaging system for breast augmentation planning. *Aesthet Surg J* 2011;**31**: 925–938.
- Losken A, Seify H, Denson DD, Paredes AA, Carlson GW. Validating three-dimensional imaging of the breast. Ann Plast Surg 2005;54:471–476.
- 14. Donfrancesco A, Montemurro P, Hedén P. Three-dimensional simulated images in breast augmentation surgery: an investigation of patients' satisfaction and the correlation between prediction and actual outcome. Plast Reconstr Surg 2013;**132**:810–822.
- Galdino GM, Nahabedian M, Chiaramonte M, Geng JZ, Klatsky S, Manson P. Clinical applications of three-dimensional photography in breast surgery. Plast Reconstr Surg 2002;110:58–70.
- del Palomar AP, Calvo B, Herrero J, López J, Doblaré M. A finite element model to accurately predict real deformations of the breast. *Med Eng Phys* 2008;**30**:1089–1097.
- Thanoon D, Garbey M, Bass BL. Deriving indicators for breast conserving surgery using finite element analysis. Comput Methods Biomech Biomed Engin 2015;18:533–544.
- Vavourakis V, Eiben B, Hipwell JH, Williams NR, Keshtgar M, Hawkes DJ. Multiscale mechano-biological finite element modelling of oncoplastic breast surgery – numerical study towards surgical planning and cosmetic outcome prediction. PloS One 2016;11:e0159766.
- O'Connell RL, DiMicco R, Khabra K, O'Flynn EA, deSouza N, Roche N et al. Initial experience of the BREAST-Q breast-conserving therapy module. Breast Cancer Res Treat 2016;160:79–89.
- 20. Pusic AL. Measuring quality of life in breast surgery: content development of a new modular system to capture patient-

reported outcomes (The MSKCC Breast-Q). In: ISOQOL Annual Meeting, Lisbon, Portugal 2006. https://www.isoqol.org/events/.

- O'Connell RL, Di Micco R, Khabra K, Wolf L, deSouza N, Roche N et al. The potential role of three-dimensional surface imaging as a tool to evaluate aesthetic outcome after Breast Conserving Therapy (BCT). Breast Cancer Res Treat 2017;164:385–393.
- Singh P, Pearlman S. Use of computer imaging in rhinoplasty: a survey of the practices of facial plastic surgeons. Aesthetic Plast Surg 2017;41:898–904.
- Persing S, Timberlake A, Madari S, Steinbacher D. Three-dimensional imaging in rhinoplasty: a comparison of the simulated versus actual result. Aesth Plast Surg 2018;42:1331–1335.
- Lekakis G, Hens G, Claes P, Hellings PW. Three-dimensional morphing and its added value in the rhinoplasty consult. Plast Reconstr Surg Glob Open 2019;7:e2063.
- 25. Yip JM, Mouratova N, Jeffery RM, Veitch DE, Woodman RJ, Dean NR. Accurate assessment of breast volume: a study comparing the volumetric gold standard (direct water displacement measurement of mastectomy specimen) with a 3D laser scanning technique. Ann Plast Surg 2012;68:135–141.
- de Runz A, Boccara D, Bertheuil N, Claudot F, Brix M, Simon E. Three-dimensional imaging, an important factor of decision in breast augmentation. Ann Chir Plast Esthet 2018;63:134–139.
- Regnault P. Breast ptosis. Definition and treatment. Clin Plast Surg 1976;3:193-203.
- Knoops PGM, Papaioannou A, Borghi A, Breakey RWF, Wilson AT, Jeelani O et al. A machine learning framework for automated diagnosis and computer-assisted planning in plastic and reconstructive surgery. Sci Rep 2019;9:13597.
- Friedman AJ, Cosby R, Boyko S, Hatton-Bauer J, Turnbull G. Effective teaching strategies and methods of delivery for patient education: a systematic review and practice guideline recommendations. J Cancer Educ 2011;26:12–21.
- Waller A, Forshaw K, Bryant J, Carey M, Boyes A, Sanson-Fisher R. Preparatory education for cancer patients undergoing surgery: a systematic review of volume and quality of research output over time. Patient Educ Couns 2015;23:S0738–3991.
- Cho HSM, Davis GC, Paek JE, Rao R, Zhao H, Xie X-J et al. A randomised trial of nursing interventions supporting recovery of the postmastectomy patient. J Clin Nurs 2013;22:919–929.
- Maguire P, Brooke M, Tait A, Thomas C, Sellwood R. The effect of counselling on physical disability and social recovery after mastectomy. Clin Oncol 1983;9:319–324.
- Fallowfield LJ, Hall A, Maguire P, Baum M, A'Hern RP. A question of choice: results of a prospective 3-year follow-up study of women with breast cancer. *Breast* 1994;3:202–208.
- Preminger BA, Lemaine V, Sulimanoff I, Pusic AL, McCarthy CM. Preoperative patient education for breast reconstruction: a systematic review of the literature. J Cancer Educ 2011;26:270–276.
- Contant CM, van Wersch AM, Wiggers T, Wai RT, van Geel AN. Motivations, satisfaction, and information of immediate breast reconstruction following mastectomy. *Patient Educ Couns* 2000; 40:201–208.
- Sheehan J, Sherman KA, Lam T, Boyages J. Association of information satisfaction, psychological distress and monitoring coping style with post-decision regret following breast reconstruction. Psychooncology 2007;16:342–351.



# **European Colorectal Congress**

28 November – 1 December 2022, St.Gallen, Switzerland

#### Monday, 28 November 2022

09.50 **Opening and welcome** Jochen Lange, St.Gallen, CH

10.00 It is leaking! Approaches to salvaging an anastomosis Willem Bemelman, Amsterdam, NL

10.30 Predictive and diagnostic markers of anastomotic leak Andre D'Hoore, Leuven, BE

11.00 SATELLITE SYMPOSIUM

**ETHICON** 

11.45 Of microbes and men - the unspoken story of anastomotic leakage James Kinross, London, UK

#### 1215 LUNCH

13.45 **Operative techniques to reduce** anastomotic recurrence in Crohn's disease Laura Hancock, Manchester, UK

14.15 Innovative approaches in the treatment of complex Crohn Diseases perianal fistula Christianne Buskens, Amsterdam, NL

14.45 To divert or not to divert in Crohn surgery technical aspects and patient factors Pär Myrelid, Linköping, SE

15.15 **COFFEE BREAK** 

15.45 Appendiceal neoplasia - when to opt for a minimal approach, when and how to go for a maximal treatment Tom Cecil, Basingstoke, Hampshire, UK

#### 16.15 **SATELLITE SYMPOSIUM Medtronic**

17.00 **Outcomes of modern induction therapies** and Wait and Watch strategies, Hope or Hype Antonino Spinelli, Milano, IT

17.30 **EAES Presidential Lecture - Use of ICG in** colorectal surgery: beyond bowel perfusion Salvador Morales-Conde, Sevilla, ES



18.00 **Get-Together with your colleagues** Industrial Exhibition

9.00 **CONSULTANT'S CORNER** Michel Adamina, Winterthur, CH

10.30 **COFFEE BREAK** 

11 00 SATELLITE SYMPOSIUM INTUITIVE

11.45 Trends in colorectal oncology and clinical insights for the near future Rob Glynne-Jones, London, UK

12.15 LUNCH

1345 **VIDEO SESSION** 

14.15 **SATELLITE SYMPOSIUM** 

### BD

15.00 **COFFEE BREAK** 

15.30 The unsolved issue of TME: open, robotic, transanal, or laparoscopic shining light on evidence and practice Des Winter, Dublin, IE Jim Khan, London, UK Brendan Moran, Basingstoke, UK

16.30 SATELLITE SYMPOSIUM

Takeda



1715 **Lars Pahlman lecture** Søren Laurberg, Aarhus, DK

Thursday, 1 December 2022 Masterclass in Colorectal Surgery Proctology Day

#### Wednesday, 30 November 2022

9 00 Advanced risk stratification in colorectal cancer - choosing wisely surgery and adjuvant therapy Philip Quirke, Leeds, UK

09.30 **Predictors for Postoperative Complications** and Mortality Ronan O'Connell, Dublin, IE

10.00 Segmental colectomy versus extended colectomy for complex cancer Quentin Denost, Bordeaux, FR

10.30 **COFFEE BREAK** 

11 00 Incidental cancer in polyp - completion surgery or endoscopy treatment alone? Laura Beyer-Berjot, Marseille, FR

11 30 SATELLITE SYMPOSIUM

12.00

Less is more – pushing the boundaries of full-thickness rectal resection Xavier Serra-Aracil, Barcelona, ES

12 30 LUNCH

14.00 **Management of intestinal** neuroendocrine neoplasia Frédéric Ris, Geneva, CH

14.30 **Poster Presentation & Best Poster Award** Michel Adamina, Winterthur, CH

15.00 **SATELLITE SYMPOSIUM OLYMPUS** 

15.45 **COFFEE BREAK** 

16.15 **Reoperative pelvic floor surgery –** dealing with perineal hernia, reoperations, and complex reconstructions Guillaume Meurette, Nantes, FR

16.45 Salvage strategies for rectal neoplasia Roel Hompes, Amsterdam, NL

**Beyond TME – technique and results** of pelvic exenteration and sacrectomy Paris Tekkis, London, UK

19.30 **FESTIVE EVENING** 

# Information & Registration www.colorectalsurgery.eu